Author's response to reviews

Title: Hyponatremic Hypertensive Syndrome (HHS) in an 18-Month Old-Child Presenting as Malignant Hypertension: Case Report

Authors:

Dr Mehul P Dixit (mdixit@peds.arizona.edu)
John D Hughes (jhughes@email.arizona.edu)
Andreas Theodorou (aat@peds.arizona.edu)
Naznin M Dixit (ndixit@peds.arizona.edu)

Version: 5 Date: 3 Apr 2004

Title - removed all unnecessary capitalisation from the title.
* Affiliations - Each affiliation is written out separately, in full and in the following format: department, institute, city, country. Please ensure that the country is given for both affiliations.
* Section headings - removed all unnecessary capitalisation.
* Discussion - section heading was ommited.
* Conclusion - renamed this section 'Conclusions'.
* Authors' contribution - renamed this section 'Authors' contributions'.

Acknowledgements: publication "of" the study